# UC San Diego UC San Diego Previously Published Works

# Title

Comparison of eating disorders symptoms and body image between individual and team sport adolescent athletes during the COVID-19 pandemic

# Permalink

https://escholarship.org/uc/item/6zx100gp

**Journal** Journal of Eating Disorders, 10(1)

**ISSN** 2050-2974

# Authors

Firoozjah, Morteza Homayounnia Shahrbanian, Shahnaz Homayouni, Alireza <u>et al.</u>

# Publication Date

2022

# DOI

10.1186/s40337-022-00644-4

Peer reviewed

**Open Access** 

# Comparison of eating disorders symptoms and body image between individual and team sport adolescent athletes during the COVID-19 pandemic

Morteza Homayounnia Firoozjah<sup>1</sup>, Shahnaz Shahrbanian<sup>2</sup>, Alireza Homayouni<sup>3</sup> and Heather Hower<sup>4,5\*</sup>

# Abstract

**Background:** COVID-19 has significantly disrupted the routines of school sports for adolescent athletes, which can affect their usual eating behaviors and body image. Specific pressures of individual sports (which tend to emphasize "leanness" as a means to improving performance), versus team sports (which tend to not require "leanness" for an athlete to be competitive), may further increase the risk of disordered eating (DE), eating disorders (ED), and distorted body image. An additional factor to consider is the gender of the athletes, with participation in "lean" sports associated with increased DE and body dissatisfaction for male, but not female, athletes.

**Methods:** Participants of the study included 124 Iranian male adolescent athletes residing in Mazandaran province (one of the most affected areas of Iran during COVID-19), who played in 1 of 6 sports (3 individual, 3 team). ED symptoms were assessed by the Eating Attitudes Test-26 (EAT-26), and body image was assessed by the Body-Esteem Scale for Adolescents and Adults (BESAA).

**Results:** The individual athlete group (n = 62) had significantly higher EAT-26 subscale scores for Bulimia and Food Preoccupation (p = 0.019), as well as significantly higher BESAA subscale scores for Appearance (p = 0.001), Weight (p = 0.001), and Attribution (p = 0.001), compared to the team athlete group (n = 62). However, there were no significant differences between the two athlete groups on the EAT-26 Dieting and Oral Control subscales.

**Conclusions:** COVID-19 presents specialized issues for adolescent athletes, particularly those at risk for, or experiencing, DE, ED, and distorted body image. While individual athletes had significantly higher scores than team athletes on most subscales, there were no differences between groups on subscales of dieting and oral control. Overall, the findings highlight the need for sport psychologists, coaches, and other sports professionals working with male adolescent athletes (individual and team) to monitor DE, ED, and distorted body image during COVID-19, in order to provide early intervention, and mitigate the risk of long-term consequences.

**Plain English Summary:** COVID-19 has significantly disrupted the routines of school sports for adolescent athletes, which can affect their usual eating behaviors and body image. Specific pressures of individual sports (which tend to emphasize "leanness" as a means to improving performance), versus team sports (which tend to not require "leanness"

\*Correspondence: Heather\_Hower@Brown.edu

<sup>4</sup> Department of Psychiatry, Eating Disorders Center for Treatment and Research, University of California at San Diego School of Medicine, 4510 Executive Drive, San Diego, CA 92121, USA Full list of author information is available at the end of the article



© The Author(s) 2022. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/ficenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

for an athlete to be competitive), may further increase the risk of disordered eating (DE), eating disorders (ED), and distorted body image. An additional factor to consider is the gender of the athletes, with participation in "lean" sports associated with increased DE and body dissatisfaction for male, but not female, athletes. Participants of the study included 124 Iranian male adolescent athletes in Mazandaran province (one of the most affected areas of Iran during COVID-19), who played in 1 of 6 sports (3 individual, 3 team). The individual athlete group (62 participants) had significantly higher scores on measures of ED (Bulimia, Food Preoccupation), and body image (Appearance, Weight, Attribution), versus the team athlete group (62 participants). However, there were no significant differences between groups on other measures of ED (Dieting, Oral Control). Findings highlight the need for sport professionals working with adolescent athletes to monitor DE and body image during COVID-19 for early intervention.

Keywords: Eating disorders, Body image, Adolescent athletes, COVID-19 pandemic

# Background

# COVID-19 general population lowered physical activity and increased psychological distress

The global outbreak of COVID-19, from March 2020 to date, has resulted in closure of multiple recreational facilities (e.g., gyms) around the world. Many individuals are not able to participate in their regular individual or group physical activities. According to the World Health Organization (WHO), under such conditions, many people tend to be less physically active, have longer screen time, irregular sleep patterns, as well as reduced quality of diet, potentially resulting in weight gain, and loss of physical fitness, which can have a distinct impact on athletes [1]. COVID-19 has also been associated with significantly higher levels of psychological distress, including elevated rates of anxiety and depression, which can have a negative effect on people's functioning in different domains, overall well-being, identity, and quality of life [2]. Indeed, there was an early (April 2020) call to action for multidisciplinary, collaborative, mental health research conducted within populations whose lives are impacted by the pandemic [3].

# COVID-19 adolescent athlete lowered physical activity and increased psychological distress

Research on the specific population of adolescent athletes with lowered physical activity (e.g., restricted exercise routines) and increased psychological distress (e.g., exacerbated eating and body image concerns) during COVID-19 is very limited, given the narrow focus of this study group, and the acute onset of the pandemic. With this caveat in mind, we can learn from existing studies on aspects of this population, and dependent variables of interest, in order to compare and contrast findings with the current study. Overall, the existing literature thus far indicates that adolescent athletes have reported decreased physical activity, increased depression and anxiety symptoms, and lower quality of life, during the COVID-19 pandemic [4, 5]. Near the beginning of the pandemic (May 2020), McGuine et al. compared changes in health measures of adolescent athletes in a cohort of participants prior to the pandemic (2015–2018) to a cohort of participants during the pandemic (May 2020) [4]. They reported that the adolescent athletes' scores for physical activity, depression symptoms, and quality of life were lower during COVID-19, compared to years earlier [4]. The authors reaffirm that quantifying these types of changes in subsequent research studies will allow health care providers to implement strategies to improve the health of adolescent athletes during the pandemic [6].

Denerel and colleagues assessed the effects of longduration home isolation linked to the pandemic on the mental health of adolescent athletes in June 2021 [5]. In a study that included 940 team athletes, 274 individual athletes, and 131 nonathlete controls (both boys and girls, age range 12-17 years), they reported that 88% did not meet the physical activity recommendations for children and adolescents, and 42.8% felt depressed. Among athletes, depressive and anxiety symptoms were lower compared with nonathlete controls (p < 0.01). Posttraumatic stress symptoms were lower among athletes than nonathlete controls for girls (team vs. control, p = 0.006; individual vs. control, p = 0.002) but similar for boys (p > 0.05). The depression (p=0.518), state (p=0.866), and trait anxiety (p = 0.507) symptoms were similar between team athletes and individual athletes. The authors reaffirmed that, though adolescent athletes' depression, anxiety, and posttraumatic stress symptoms were significantly lower than nonathlete controls, athletes also had high depression levels, indicating the need to take precautions to protect the psychological health of athletes in the pandemic [5].

## COVID-19 disruption of adolescent athlete routines

COVID-19 has significantly disrupted the routines of school and sports for adolescents (e.g., in-person training activities, organized sporting events), which can affect their usual eating behaviors, body image, and exercise patterns [4, 7–12]. The increased focus on body weight during adolescence can lead to disordered eating (DE), eating disorders (ED), and body image concerns [13]. Isolation, including during the COVID-19 lockdown, can additionally challenge an adolescent athlete's ability to maintain their usual healthy nutrition and performance levels [14]. Athletes, used to a high caloric intake, may maintain the same eating routines, despite less exercise, and evolving mood-related drivers of comfort eating occurring with boredom and stress [8]. Some of the beneficial physical adaptations in the body observed as a result of a regimented training program (e.g., increases in blood and blood plasma volume, cardiac output and stroke volume during maximal efforts, muscular hypertrophy) [15] have been shown to reverse with deconditioning (e.g., extended periods of lowered or inactivity) [16]. Thus, adolescent athletes may be particularly vulnerable to these concerns during this extended COVID-19 lockdown [3–12].

#### The DE to ED spectrum

The DE to ED spectrum ranges from the DE subclinical state, characterized with less cognitive and behavioral symptomatology [9, 17], to the clinical ED diagnoses, as defined by the Diagnostic and Statistical Manual of Mental Disorders-5 (DSM-5) [18]. The DSM-5 includes the primary diagnoses of Anorexia Nervosa (AN), characterized by restriction and low weight, and Bulimia Nervosa (BN), characterized by recurrent episodes of binging and purging. ED are often contributed to by a combination of genetic predisposition (family history) and environmental triggers (e.g., sport-specific pressures, COVD-19 stress). ED are associated with certain risk factors which may be especially developmentally relevant for adolescent athletes. For most ED, the age of onset is usually in adolescence, and the consequential effects on the developing body can significantly interfere with an adolescent's athletic performance. Indeed, research indicates that individuals with a family history of AN, who have childhood anxiety traits, and pursue avocations that encourage thinness (e.g., elite athletes), are at an increased risk for developing AN in adolescence [18]. Similarly, individuals with a family history of BN, who have childhood obesity, subsequently attempted to restrict food and/or overexercise (e.g., through sports), but ended up binging later, are at increased risk for developing BN in adolescence [18].

#### **Body image**

A cross-cutting symptom of the above primary types of ED is the concern about body image, which is the internal representation of an individual's external appearance (e.g., self-evaluations of one's physical appearance) [19]. Given that appearance is an integral part of one's identity, and it plays a role in social situations, body image is an important aspect of life for most people, and particularly for those individuals who are at risk for, or are experiencing, an ED [20].

Formative research on body image in adolescent male athletes has indicated that there are significant differences between the types of sports that athletes participate in, and their related body image concerns [21]. A study by Parks and Read [21] compared adolescent male football players (team sport) (n=44) and crosscountry runners (individual sport) (n=30) on measures of weight concerns, and perceived/ideal body shape/size. They noted that football players reported a more positive body image, while cross-country runners indicated a greater degree of body dissatisfaction, and concern for weight control [21]. The authors concluded that the cross-country runners' group should be identified for increased need in health education [21].

Emerging evidence (2021) suggests that public health messaging associated with COVID-19 and increased reliance on videoconferencing technologies have had a negative impact on body image [22, 23]. A recent (March 2022) mixed-studies age-inclusive systematic review of the experiences of body image, DE, and ED during the pandemic by Schneider and colleagues [24] reported that, overall, studies have shown increased shape and weight concerns [25], drive for thinness/ muscularity [26], and decreased self-esteem [27].

Research specifically conducted on adolescents and body image in May 2020 found a significant association between the frequency of following appearancefocused accounts on social media (Instagram) and an increase in body and appearance dissatisfaction [28]. Another study conducted August–December 2020 by Lessard and Puhl reaffirmed that COVID-19 increases in body dissatisfaction were prevalent, with the majority of participants reporting increased exposure to weight stigmatizing social media content during the pandemic [29].

In an early pandemic mixed methods study (conducted April–May 2020, published in 2021), Buckley and colleagues collected data on body image transitions of current and former adult athletes. They noted that 34.8% (n=69) of participants self-reported worsened body images, which were directly attributed to COVID-19 (increased body preoccupation, and fear of body composition changes) [8]. They reaffirmed that body image is one component of athletic identity; the strength with which people identify with, and embrace, their role as an athlete, including the aesthetic body image of their sport [30]. When there are changes (e.g., COVID-19 athletic disruptions, decrease in physical fitness, increase in

weight), athletes can experience a sense of identity loss that increases the risk for depression [31].

To the best of these authors' knowledge, research specifically on COVID-19 adolescent athlete changes in body image has yet to be conducted. However, based upon the above noted studies [20-28, 30-32], we can speculate that there will be a worsening of athletes' body image over the duration of the pandemic.

## COVID-19 exacerbation of ED behavioral symptoms

Research on the exacerbations of ED during the pandemic have included initial reports (April 2020) and systematic reviews (2021-to date) among adult and adolescent populations. Touyz and colleagues provided one of the first age-inclusive perspectives on this topic (April 2020) [33]. They noted that, for many individuals with AN, lack of access to the usual forms of exercise (e.g., going to the gym) heightened concerns about gaining weight, leading to further restriction and emaciation, which increased their potential for experiencing more serious COVID-19 complications [33]. Similarly, for many individuals with BN, increased access to food while in guarantine at home, without the normal athletic exercise routine to balance out the calorie intake, resulted in reported weight gain that they feared may hinder their current and future athletic performance [33].

Buckley et al., in their study of current and former adult athletes, reported a surge in DE, including 21.1% of participants with a diagnosable ED, and 32.8% with selfreported worsened food relationships during this early phase of the pandemic [8]. Qualitative analysis indicated that DE occurred in the form of inhibitory food control, and binge eating [8]. In a subsequent scoping review of the COVID-19 ED research (November 2021), Linardon et al. reported that, in the 70 (mostly adult) selected studies, those most susceptible to pandemic symptom escalation were individuals with confirmed ED, as well as those in at-risk populations, including athletes [34].

The early impact of COVID-19 on adolescents with ED indicated increases in symptoms, impairment, and need for immediate treatment. Graell et al. reported higher reactivation of ED symptoms in adolescents compared to children from March–May 2020 [35]. During a similar time period (April–October 2020), Spettigue et al. compared a cohort of adolescent participants who presented for ED assessment before the pandemic (2019) with those who presented during the pandemic (2020) [36]. They found that pandemic cohort trended towards having lower percentage of goal weights, as well as higher rates of self-reported impairment, and were significantly more likely to be medically unstable (p=0.005), and to require hospitalization (p=0.005) [36]. Higher rates of inpatient admissions, emergency room consultation requests,

and outpatient referrals deemed "urgent" were likewise associated with the pandemic period [36]. A very recent meta-analysis conducted by Haghshomar et al. (April 2022) focused on 13 studies with age ranges of 13–70, indicating that the overall pooled prevalence of exacerbation of binge eating, food restriction, purging behaviors, and concerns about food intake, in the sample of 7848 was 59.65% (95% CI: 49.30%; 69.60%) [37].

Similar to the body image research noted above, to the best of these authors' knowledge, studies specifically on COVID-19 adolescent athlete exacerbation of ED behavioral symptoms have yet to be conducted. Nevertheless, the existing literature [34–38] suggests a worsening of their symptoms as the pandemic continues.

## Sport-specific pressures

Another aspect of an athlete's identity includes how they respond to the unique pressures of their specific sport. Petrie and Greenleaf formulated a theoretical etiological model of eight mediating factors in the development of DE in athletes [39, 40]. One factor is the type of sport (individual vs. team sports, and what they termed as "lean vs. non-lean sports"). They note that individual sports tend to be ones that emphasize "leanness" [41], with the belief that lower body weight improves performance (e.g., track and field, martial arts) [41]. Alternatively, team sports tend to be ones that are "non-lean" [42], in that they do not require a low body weight in order for an athlete to be competitive (e.g., basketball, soccer, volleyball) [42]. "Lean" sports may increase risk for DE, in part through a syndrome known as Relative Energy Deficiency in Sport (RED-S) [43]. This involves impaired physiological function including, but not limited to, metabolic rate, menstrual function, bone health, immunity, protein synthesis, and cardiovascular health, caused by relative energy deficiency, which can affect male as well as female athletes. A recent systematic review [44] of the prevalence of DE in athletes categorized by emphasis on "leanness" and activity reported that six out of the seven included studies found a significant increase in DE rates among "lean" sport types, compared to "non-lean" sport types [44].

Sports can be further divided into six subgroups: aesthetic, weight-dependent, endurance, ball game, power, and technical sports [45]. Of these categories, aesthetic, weight-dependent, and endurance sports are typically considered "lean" sports (as defined by Petrie and Greenleaf, [39, 40]), whereas ball game, power, and technical sports are considered "non-lean" sports [39, 40]. In particular, aesthetic sports, which are judged by a complex set of rules favoring appearance (e.g., ballet, gymnastics), have been shown in sport-specific studies to have a higher prevalence of DE and ED compared to other sport subgroups [46–48]. Weight-dependent sports, which divide competitors into different categories based on their weight (e.g., wrestling, martial arts), have similarly been shown to have a higher ED pathology than those in other sport subgroups [49, 50]. Endurance sports typically associate lower body weight with a higher level of competition (e.g., track and field-running, cross-country skiing) [51].

Another sport-specific factor to consider is peer competition, which can exacerbate DE, ED, and distorted body image [52]. While both team and individual sports require competition, cooperation is much more prevalent in team than in individual sports. Thus, it is possible that team athletes' additional goal of cooperation more evenly balances out the goal of competition, compared to individual athletes' sole requirement for competition [53]. Additionally, peer competition can lead athletes to deny the seriousness of their concerns, and if they do seek help from a doctor, it is more likely due to complaints of decreased performance, rather than the symptoms of an ED, per se [53].

#### Gender-specific pressures

Although EDs are often thought of as feminine illnesses, epidemiological studies indicate that males are also atrisk for developing ED [54–56]. Most notably, adolescent males may be less likely to seek treatment than females due to an overall higher degree of shame and stigma related to their ED [57–59]. Further, physicians and other health care providers may be less likely to recognize DE symptoms in adolescent males due differences in symptom presentation (e.g., male focus on "leanness," vs. female focus on weight) [58, 59].

These gender-specific pressures can overlap with the above-noted sport-specific pressures. A systematic review [60] of risk factors for eating psychopathology included a study which examined the relationship between gender, type of sport ("lean" vs. "non-lean"), body dissatisfaction, and self-esteem, with DE behaviors in Division 1 college athletes [60]. They found that participating in "lean" sports was associated with increased DE and body dissatisfaction for male athletes, but not female athletes [61]. This unique vulnerability for male athletes, in addition to the factors related to lower likelihood of treatment for males (e.g., stigma, shame, provider misconceptions), suggests that male athletes should be targeted by sport psychologists, coaches, and other sports professionals, for prevention and early intervention for DE, ED, and distorted body image [62].

### Aim of the present study

COVID-19 thus presents specialized issues for adolescent athletes, particularly those at risk for, or experiencing,

DE, ED, and associated distorted body image [3, 5–12]. Specific pressures of individual sports (which tend to emphasize "leanness" as a means to improving performance), versus team sports (which tend to not require "leanness" for an athlete to be competitive), may exacerbate these concerns. An additional factor to consider is the gender of the athletes, with participation in "lean" sports associated with increased DE and body dissatisfaction for male, but not female, athletes. The aim of the present study was therefore to compare ED symptoms and body image between individual and team sport adolescent male athletes living in one of the most affected areas of Iran at the beginning of COVID-19.

# Methods

## Participants

Participants of the study included 124 Iranian male adolescent athletes residing in the Mazandaran province (one of the most affected areas of Iran during COVID-19), who were recruited through the three University study sites nearby (Farhangian University, Tarbiat Modares University, and Islamic Azad University). The athletes played in 1 of 6 sports (individual or group): (1) Taekwondo (individual); (2) Track and Field (individual); (3) Wrestling (individual); (4) Volleyball (group); (5) Basketball (group); or (6) Soccer (group).

Study inclusion criteria were: (1) aged 12 to 19 years old ("adolescence," as defined by Rice, F.P., in Human Development: A Life-Span Approach) [63]; (2) completion of a physical illness history measure, assessed pre-study through participant self-report via the WHO Global Physical Activity Questionnaire (WHO-GPAQ) [64] (no disabilities or physical problems were reported by the research participants); (3) completion of a mental illness history measure, assessed pre-study through participant self-report via the WHO-Composite International Diagnostic Interview (WHO-CIDI) [65] (no specific, diagnosable mental illnesses were reported by the research participants); (4) sports history for at least 3 years, assessed via the WHO-GPAQ; and (5) written consent by the adolescents and their parents for participation in the research. The study exclusion criterion was: (1) lack of regular physical activity, assessed via the WHO-GPAQ. This criterion was included in order to differentiate between a lack of regular physical activity due to normal patterns (e.g., sedentary lifestyle), versus a lack of regular physical activity due to changes during the COVID-19 pandemic assessment period (e.g., reduced exercise levels). Of note, all included participants had continued exercise during the pandemic (see details below). The amount of regular physical activity was tracked by both the adolescent athletes and their parents, and the accuracy of the level of activity was confirmed by the study

researchers. Thus, all participant responses were equal in terms of the associations between activity and scores.

Of the eligible participants who were available during a COVID-19 quarantine period from June through August 2020 (N=180), the Krejcie and Morgan Sampling Method was used to simplify the process of determining the sample size for a finite population [66], resulting in a calculation of N=124 sample participants. In order to maximize statistical power, the eligible participants were then selected to be divided equally into two groups based upon their sport affiliation: the individual sport group (n=62), and the team sport group (n=62). Throughout the data collection quarantine time, all athletes practiced and played at home, or in socially isolated environments, depending on the training that the coaches designed for them. Thus, the athletes had continued exercise during the lockdown, albeit with modified routines and formats.

### Procedure

The governing Medical Ethics and History of Medicine Research Center for each Iranian university study site reviewed and approved the study before enrollment. Participants were recruited via a study research coordinator who had contacted the respective officials of the 6 sports groups. The study procedures were explained, and informed consent was obtained from all participants and their parents prior to study initiation.

#### Measures

De-identified study data (sociodemographic variables, questionnaires) were collected online at the participants' homes during the quarantine period of COVID-19. The Persian (Farsi) translations of the questionnaires were used for the participants' comfort and familiarity with the language. These translated versions were evaluated, and found to be reliable and valid for the Iranian population (please see below for the psychometrics of each questionnaire).

#### WHO-GPAQ

The WHO-GPAQ collects information on physical activity participation in three domains (activity at work, travel to and from places, recreational activities), as well as sedentary behavior, defined as "any waking behavior characterized by a low energy expenditure (e.g., sitting, as with desk work, driving a car, watching television)" [64]. Participants completed the self-report version pre-study to denote their histories of physical illness and sports participation. In a nine country reliability and validity study of the WHO-GPAQ, Bull and colleagues reported that the inter-rater and test–retest reliability coefficients were of moderate to substantial strength (Kappa k=0.67-0.73; Spearman's rho  $\rho=0.67-0.81$ , p<0.01, respectively) [67]. Results on concurrent validity between the WHO-GPAQ and the International Physical Activity Questionnaire (IPAQ), a previously validated and accepted measure of physical activity [68], also showed a moderate to strong positive relationship (Pearson's product-moment correlation coefficient r = 0.45 - 0.65 [67]. In the present study, the Persian (Farsi) translation version of the whole scale WHO-GPAQ had an internal consistency reliability Cronbach's alpha  $\alpha = 0.93$  [69]. Utilizing Lawshe's technique to assess content validity [70], subject matter experts review each measurement question in order to determine its individual content validity ratio (CVR), with a final mean CVR of all of the questions denoted as the content validity index (CVI) (the closer the CVI is to 1, the higher the overall content validity of the measure). In this translation version, the CVI = 0.72 [69].

# WHO-CIDI

The WHO-CIDI includes a screening module and 41 sections that focus on mental health diagnoses, functioning, treatment, risk factors, socio-demographic correlates, and different methodological factors [65]. The WHO-CIDI generates diagnoses of mental disorders according to the criteria of the DSM-5 [18]. Participants completed the self-report version pre-study to denote their history of mental illness. In a study of the reliability, validity, and factorial structure of the WHO-CIDI in Iranian psychiatric outpatients, Dadfar and Kalibatseva reported an internal consistency reliability Cronbach's alpha  $\alpha = 0.91$  [71]. The WHO-CIDI negatively correlated with the Patient Health Questionnaire-9 (PHQ-9) [72] (Pearson's product moment correlation coefficient r = -0.358), Patient Health Questionnaire-15 (PHQ-15) [73] (Pearson's product moment correlation coefficient r = -0.328), and the Beck Depression Inventory-13 (BDI-13) [74] (Pearson's product moment correlation coefficient r = -0.475), indicating good validity [71]. In the present study, the Persian (Farsi) translation version of the whole scale WHO-CIDI had an internal consistency reliability Cronbach's alpha  $\alpha = 0.83$ , and CVI = 0.72 [75].

#### Eating attitudes test-26 (EAT-26)

The EAT-26 [76] is a 26-item self-report measure with 3 subscales: (1) thoughts and behaviors related to dieting (Dieting); (2) preoccupation with food and impulses to binge and purge (Bulimia and Food Preoccupation); and (3) attempts to control food intake (Oral Control). Response agreement is rated on a 4-point Likert scale (0="Never," to 3="Always"), with total scores ranging from 0 to 78 (clinical cut-off score of  $\geq$  20 indicates disordered eating). Garner and colleagues reported that the wholescale EAT-26 score items have a "good" [77] internal consistency reliability Cronbach's alpha  $\alpha = 0.88$ , and the Dieting subscale items have an "excellent" [77] internal consistency reliability Cronbach's alpha  $\alpha = 0.90$ . While the other two internal consistency reliabilities were not initially reported, a subsequent meta analyses of the EAT-26 literature by Gleaves et al. [78] reported a mean "questionable" [77] internal consistency reliability Cronbach's alpha  $\alpha = 0.67$  for the Bulimia and Food Preoccupation subscale, and a mean "poor" [77] Cronbach's alpha  $\alpha = 0.56$  for the Oral Control subscale. However, the wholescale EAT-26 score items have a content validity Pearson's product moment correlation r = 0.91, and the 3 subscales have a content validity Pearson's product moment correlation r = 0.78[76].

Research findings have indicated that the wholescale EAT-26 score items in general populations and patient samples have been shown to be highly reliable (internal consistency reliability Cronbach's alpha  $\alpha$ =0.91; test-retest reliability Spearman's rho  $\rho$ =0.98, p<0.01), and valid (positively correlated with the Structured Clinical Interview for Axis I DSM-IV Disorders (SCID) for Module H [79]; criterion validity Pearson's product moment correlation r=0.90) [80–82]. In the present study, the Persian (Farsi) translation version of the wholescale EAT-26 had an internal consistency reliability Cronbach's alpha  $\alpha$ =0.90, and the test-retest reliability for the EAT-26 3 subscales was good (Spearman's rho  $\rho$ =0.84–0.89, p<0.001) [83].

#### Body-esteem scale for adolescents and adults (BESAA)

The BESAA [84] is a 23-item self-report measure with 3 subscales: (1) feelings regarding one's appearance (Appearance); (2) perceptions of others' evaluations about one's appearance (Attribution); and (3) weight satisfaction (Weight). Response agreement is rated on a 5-point Likert scale (1="Never," to 5="Always"), with total scores ranging from 23 to 115 (higher scores indicate more positive body-esteem). Mendelson and colleagues reported that the BESAA items have an internal consistency reliability Chronbach's alpha  $\alpha = 0.89$ , Spearman's rho  $\rho = 0.88$ , p < 0.01, and a content validity Pearson's product moment correlation r = 0.72 [84]. In the present study, the Persian (Farsi) translation version of the BESSA demonstrated good model fit statistics  $(\chi^2/df = 3.41, p < 0.001)$ , and good internal consistency reliability for the 3 subscales: (1) BESSA-Appearance (Chronbach's alpha  $\alpha = 0.76$ ), (2) BESSA-Attribution (Chronbach's alpha  $\alpha = 0.69$ ), and (3) BESSA-Weight (internal consistency reliability Chronbach's alpha  $\alpha = 0.85$ ) [85]. The test–retest reliability for the BESSA 3 subscales was moderate (Spearman's rho  $\rho = 0.57 - 0.68$ , p < 0.01 [85].

### Data analyses

All data were entered into the Statistical Package for Social Science (SPSS) 24 software for analyses [86]. Sociodemographic variables were collected for the individual and team athlete groups. The Kolmogorov-Smirnov Test [87] was used to check the normality of the EAT-26 and BESSA data (normality is the first statistical assumption required for the subsequent Multiple Analysis of Variance (MANOVA)) [88]. Of note, MANOVA was chosen instead of a single *t*-test with multiple dependent variables in order to limit the joint error rate. The M-Box Test of Equality of Covariance Matrices [89] was used to check for the equality of covariance matrices for the overall differences in EAT-26 and BESSA scores between the individual and team athlete groups (equality of covariance matrices is the second statistical assumption required for the subsequent MANOVA). Levene's Test of Homogeneity of Variance [90] was used to check the homogeneity of variance for the EAT-26 and BESSA specific subscale scores of the individual and team athlete groups (homogeneity of variance is the third statistical assumption required for the subsequent MANOVA). With the above statistical assumptions being met, the MANOVA detailed the specific dependent variables (individual EAT-26 and BESSA subscale scores) that were significantly different between the individual and team athlete groups.

### Results

# Individual and team athlete groups sociodemographic variables

The sociodemographic variables of the individual (n = 62)and team (n = 62) athlete groups are presented in Table 1. In summary, individual athletes had significantly higher mean values of age, height, weight, and Body Mass Index (BMI; as calculated by the Centers for Disease Control and Prevention (CDC) BMI Percentile Calculator for Child and Teen [91]) compared to team athletes (p=0.001). However, the range of distribution for each of these variables is a tight cluster of scores, as detailed below. Individual athletes had a significantly higher mean age of 14.1 years (Standard Deviation, SD = 0.6 years, range=12.2-18.4 years) compared to team athletes (mean=13.9 years, SD=0.8 years, range=12.1-18.7 years) (p = 0.001). Individual athletes had a significantly higher mean height of 157.2 cm (SD = 7.2 cm, range = 154.1 - 182.5 cm) compared to team athletes (mean = 156.7 cm, SD = 7.6 cm, range = 149.1 - 164.3 cm)(p=0.001). Individual athletes had a significantly higher mean weight of 66.1 kg (SD = 1.6 kg, range = 64.3-87.1 kg) compared to team athletes (mean = 64.7 kg, SD = 1.8 kg, range = 64.0–75.1 kg) (p = 0.001). Individual athletes had a significantly higher mean Body Mass Index

Variable	Individual athlete group (n $=$ 62)		Team athlete group (n $=$ 62)				
	Mean	SD	Mean	SD	т	<i>p</i> -value	
Age	14.1	0.6	13.9	0.8	6.9	0.001	
Height (cm)	157.2	7.2	156.7	7.6	5.7	0.001	
Weight (kg)	66.1	1.6	64.7	1.8	3.1	0.001	
BMI	21.1	2.1	20.1	2.2	2.8	0.001	

**Table 1** Sociodemographic variables of the individual and team athlete groups

SD = Standard Deviation; cm = centimeters; kg = kilograms; BMI = Body Mass Index, as calculated by the Center for Disease Control and Prevention (CDC) BMI Percentile Calculator for Child and Teen [91]

(BMI) of 21.1 (SD = 2.1, range = 20.3–22.4) compared to team athletes (mean = 20.1, SD = 2.2, range = 19.8–22.1) (p = 0.001).

Statistical tests for the assumptions of the overall MANOVA The M-Box Test of Equality of Covariance Matrices [89] for the overall differences in EAT-26 and BESSA scores between the individual and team athlete groups is shown in Additional file 1: Table S1. Levene's Test of Homogeneity of Variance [90] for the homogeneity of variance in the EAT-26 and BESSA subscale scores between the individual and team athlete groups is presented in Additional file 1: Table S2. The MANOVA for the overall differences in EAT-26 and BESSA scores of the individual and team athlete groups is noted in Additional file 1: Table S3. In summary, 49.7% of the variance related to the difference between the individual and team athlete groups is due to the interaction of all of the dependent variables.

#### MANOVA specific dependent variables

Table 2 details the specific dependent variables (individual EAT-26 and BESSA subscale scores) in the MANOVA that had significant differences between the individual and team athlete groups. In summary, the individual athlete group had significantly higher scores on the EAT-26 Bulimia and Food Preoccupation subscales (p=0.019), and significantly higher scores on the BESSA Appearance (p=0.001), Weight (p=0.001), and Attribution (p=0.001) subscales, compared to the team athlete group. However, there were no significant differences between the two athlete groups on the EAT-26 Dieting and Oral Control subscales.

# Discussion

The aim of the present study was to compare ED symptoms and body image between individual and team sport adolescent male athletes who were geographically located in one of the most affected areas of Iran during the beginning of COVID-19. In summary, the individual athlete group had significantly higher scores compared to the team athlete group on subscales focused on preoccupation with food and impulses to binge and purge, feelings regarding one's appearance, perceptions of others' evaluations about one's appearance, and weight satisfaction. However, there were no significant differences between the two athlete groups on subscales focused on thoughts and behaviors related to dieting, and attempts to control food intake.

As noted above, while research on the specific population of individual and team sport male adolescent athletes with exacerbated eating and body image distortions during COVID-19 is very limited, given the narrow focus of this study group, and the acute onset of the pandemic, we can draw from the available literature for comparisons with the present study. Overall, our results of significant differences between the individual

Table 2 MANOVA for the specific difference in EAT-26 and BESSA subscale scores of the individual and team athlete groups	

Dispersion source	Variable	Sum of squares	df	Average of squares	F	<i>p</i> -value
Group	EAT-26 dieting	8.002	1	8.002	1.102	0.296
	EAT-26 oral control	6.099	1	6.099	0.697	0.405
	EAT-26 bulimia and food preoccupation	14.842	1	14.842	5.666	0.019
	BESSA appearance	72.434	1	72.434	30.278	0.001
	BESSA weight	61.577	1	61.77	23.181	0.001
	BESSA attribution	19.828	1	19.828	24.599	0.001

MANOVA = Multiple analysis of variance [88]; EAT-26 = Eating attitudes test-26 [76]; BESSA = Body-esteem scale for adolescents and adults [84]

and team athlete groups on subscale scores is consistent with the majority of the research that was conducted pre-COVID-19 [54–56, 92].

A relatively early study by Rodriguez et al. (1999) reported that DE symptoms (e.g., diets, compulsive eating, and compensatory behaviors) were more frequent in adolescent exercisers who participated in individual sports compared with those participating in team sports [93]. Subsequent research by Milligan et al. [61] found that participating in "lean" sports was associated with increased DE and body dissatisfaction for university male athletes, but not female athletes. A 2018 review of ED in adolescent and young adult males by Limbers and colleagues indicated that athletes who participate in sports that value "leanness" (e.g., most individual sports) may experience pressure from coaches and/or teammates to engage in unhealthy weight control practices [62]. A systematic review by Stoyel et al. of risk factors for eating psychopathology in 2019, including articles published from 2000 to 2018, examined studies with EAT-26 scores [60]. They highlighted a finding from Kong and Harris [42] on adolescent and adult (age range 17-30 years) female athletes, where those who participated in "lean" sports had an increased rate of self-induced vomiting and laxative use, compared with the "non-lean" sport participants. In a study by Pamuk and colleagues on eating attitudes in 502 male and female university athletes who completed the EAT-26, conducted and published just before COVID-19 (2019-early 2020), individual athletes were found to have significantly higher wholescale scores, and on each of the Dieting, Bulimia and Food Preoccupation, and Oral Control subscale scores, compared to team athletes (p < 0.05) [92]. This was the case when analyzing the entire sample (both genders), as well as when male athletes only were selected in a subsample [92]. Mancine et al. [44] conducted a systematic review on the prevalence of DE in athletes categorized by emphasis on "leanness" and activity type (included articles were from January 2000-July 2019, and the findings were published in 2020). They reported that 6 out of the 7 included studies (males and females, age range 12-30 years old) found a significant increase in DE rates among "lean" sport types [44]. A populationbased study of DE in 10,172 adolescents participating in individual and team sports by Heradstveit et al. [94] (data collected in 2012, research published in 2020) noted that DE symptoms were associated with higher individual sports participation compared to those participating in team sports. Pre-COVID-19 research thus tends to highlight the significant DE differences between individual (mostly "lean") and team (mostly "non-lean") athletes [94, 95].

In the current study, our findings of individual athlete increased scores on the EAT-26 Bulimia and Food Preoccupation, and BESSA Appearance, Weight, and Attribution, subscales compared to team athletes are in line with the above noted research on individual, "lean" sports. Indeed, individual athlete pressure to maintain a "lean" body for their sport by any means necessary can include the behavior of purging (bulimia) [44]. In addition, dietary restriction has been shown to lead to an increase in food preoccupation (anorexia) [44]. Further, an individual athlete's identity tends to be tied more to their appearance, weight, and attribution, compared to a team athlete's identity [30].

Our results indicating no differences between the individual and team athlete groups on the EAT-26 Dieting and Oral Control subscales, however, is consistent more with emerging research on the effects of COVID-19 and exacerbation of DE, ED, and body image distortion. Buckley et al., in their study of current and former adult athletes conducted during the early transition period of the pandemic in April-May 2020, reported a surge in DE in all participants (N = 162; male and female athletes, from a range of 41 different individual and team sports) [8]. This included 21.1% of participants scoring above the EAT-26 20 point cut-off, suggesting an ED [76], 34.8% of participants self-reporting worsened body image, and 32.8% self-reporting a worsened food relationship, directly attributed to COVID-19 [8]. Qualitative analysis indicated that DE occurred predominantly in the form of body preoccupation, inhibitory food control, fear of body composition changes, and binge eating [8]. Interestingly, they found no significant differences between the individual and team sport groups, type of sporting category (endurance, antigravitational, ball sport, power, technical and aesthetic), or level of competition (club, state, national or international) [8]. Denerel et al. [5], in their subsequent study on the effects of long-duration home isolation linked to the pandemic on the mental health of adolescent athletes (June 2021), reported a decrease in physical activity (88%) and an increase in depression (42.8%) in their participants (n = 940 team athletes, n = 274 individual athletes, n = 131 nonathlete controls, boys and girls, age range 12-17 years). Posttraumatic stress symptoms were lower among athletes than nonathlete controls for girls, but similar for boys [5]. However, they found no significant differences in depression, state, and trait anxiety symptoms between the individual and team athletes [5].

The relationship between dieting (behaviors and thoughts related to dieting) and oral control (attempts to control food intake) tends to become a vicious, reinforcing cycle for many people. Behavioral attempts to control food in turn solidify cognitive control over food,

which increases food preoccupation [8]. Inhibitory control has a role in maintaining DE, including restrict-binge cycles, and more specifically in athletes, it can function to enhance performance focus (including areas where they feel that they have failed to meet their/others' expectations) [96]. This interaction between behavioral and cognitive control may further modulate stress in such an uncertain transition as the COVID-19 pandemic [8]. In order to address the clinical implications of these findings for sport psychologists, coaches, and other sports professionals working with both individual and team athletes, Buckley et al. [8] suggest some key Practice Principles that utilize athlete descriptions of improvements in their food-body relationship over a transition. Principles related to dieting and oral control include: (1) Control (channeling control outside of the body, and away from restrictive dietary practices, or rigid body controlling movement); (2) Variety (keeping a range of food and exercise open to counteract restraint and cognitive inflexibility); and (3) Adaptability (finding regularity in diet practices where possible, but ultimately being able to adapt to altered appetite needs, food environment changes, and food access). Thus, sports professionals may help individual and team athletes improve these skills.

Buckley and colleagues recommend additional Practice Principles that have been shown in previous studies to mend the relationships between athletes, their eating practices, and their body images [8]. These include: (1) Support (engaging with specialized options to navigate potential DE development, body image concerns, mental health challenges, and social comparison); (2) Connection (connecting to one's changing needs through intuitive eating or intuitive movement, and being able to adapt to the body's physical needs related to training and exercise), and (3) Acceptance (accepting that the body is not infinitely malleable, and seeking acceptance with nonjudgement through body shape, weight, and composition changes) [8].

Research on DE, ED, and body image distortion in male adolescent athletes during COVID-19 is in its relative infancy. Findings thus far have indicated that a large number of athletes have experienced a worsening of their symptoms during the pandemic, highlighting the importance of providing specialized support towards reducing body related anxiety, vicious food control cycles, and maladaptive exercise management, during such times of transition. Further longitudinal research is needed in order to determine the long-term effects on this population. Overall, these findings underscore the need for sport psychologists, coaches, and other sports professionals working with male adolescent athletes (individual and team), to monitor DE, ED, and distorted body image during COVID-19, in order to provide early intervention, and mitigate the risk of long-term consequences.

#### **Clinical implications**

In the current study, there were several important findings and clinical implications related to the athletic settings that the participants were in (individual vs. team sports). As noted above, COVID-19 presents specialized issues for adolescent athletes, particularly those at risk for, or experiencing, DE, ED, and distorted body image [3, 5-12]. While individual athletes had significantly higher scores than team athletes on most EAT-26 and BESSA subscales, there were no differences between the groups on the EAT-26 subscales of Dieting and Oral Control. The specific relationship between dieting (behaviors and thoughts related to dieting) and oral control (attempts to control food intake) tends to become a vicious, reinforcing cycle for many people. Improvement of skills related to control (channeling away from restrictive dietary practices), variety (keeping a range of food open to counteract restraint and cognitive inflexibility), and adaptability (being able to change with altered appetite needs, food environment, and access) may improve this dieting-oral control relationship among athletes. Additionally, skills related to support (engaging with specialized options to navigate DE development), connection (to one's changing needs through intuitive eating or intuitive movement), and acceptance (non-judgement of body shape, weight, and composition changes) may mend the relationships between athletes, their eating practices, and their body images. Overall, the findings highlight the need for sport psychologists, coaches, and other sports professionals working with male adolescent athletes (individual and team), to monitor DE, ED, and distorted body image during COVID-19, in order to provide early intervention, and mitigate the risk of long-term consequences.

#### Limitations

The findings of the current study must be viewed within the context of certain limitations.

First, the study utilized a sample of Iranian male adolescent athletes during the beginning of COVID-19 quarantine (June–August 2020); therefore, results cannot be generalized to other populations. Future research should endeavor to conduct similar studies in other populations, in order to compare, and potentially replicate, the results. Second, as noted above, research on the specific population of adolescent male athletes with existing/exacerbating DE/ED symptoms and body image disturbance during the pandemic is very limited given the narrow focus of this group, and the acute onset of the pandemic. However, we have attempted to draw from the existing studies on aspects of this population and dependent variables of interest in order to compare and contrast findings in the literature, and hopefully contribute to research on this particular high-risk population. Third, due to the early assessment period for this study, structured measurements of pandemic barriers (e.g., impact on physical and mental health for individuals, family, and friends) had not yet been developed. As these scales are now available, they should be utilized in conjunction with other instruments of interest. Fourth, the participant self-reports were not supported by additional investigator conducted diagnostic interviews, which could lead to underreporting of these concerns. Subsequent research should endeavor to also include investigator interviews (e.g., by tele-interview during COVID-19) to support any selfreport measures. Fifth, we did not conduct sport-specific analyses due to concerns of low statistical power. It is possible that the characteristics of the athletes within one or two sports may be responsible for the differences that were seen between groups, and the conclusions might thus be more specific than we presented. Future research could increase the overall and sport-specific subgroup samples in order to provide additional cell sizes for more robust analyses. Finally, the cross-sectional assessment of ED and body image in the current study preclude causal inferences for these relationships. Prospective research could employ a more experimental, longitudinal design, in order to elucidate the effects of each of these variables.

## Conclusions

COVID-19 presents specialized issues for adolescent athletes, particularly those at risk for, or experiencing, DE, ED, and distorted body image [3, 5-12]. In summary, in the current study, while individual athletes had significantly higher scores than team athletes on most EAT-26 and BESSA subscales, there were no differences between the two groups on the EAT-26 subscales of Dieting and Oral Control. Clinical implications for these findings include the improvement of skills related to control, variety, adaptability, support, connection, and acceptance, which may improve the relationships between athletes, their eating practices, and their body images) [8]. Overall, the findings highlight the need for sport psychologists, coaches, and other sports professionals working with male adolescent athletes (individual and team), to monitor DE, ED, and distorted body image during COVID-19, in order to provide early intervention, and mitigate the risk of long-term consequences.

#### Abbreviations

AN: Anorexia nervosa; BESAA: Body-esteem scale for adolescents and adults; BMI: Body mass index; BN: Bulimia nervosa; CDC: Center for disease control and prevention; cm: Centimeters; CVI: Content validity index; CVR: Content validity ratio; DSM-5: Diagnostic and statistical manual of mental disorders-5; EAT-26: Eating attitudes test-26; EBT: Evidence based treatment; ED: Eating disorders; IPAQ: International physical activity questionnaire; kg: Kilograms; MANOVA: Multiple analysis of variance; SD: Standard deviation; SPSS: Statistical package for social science; WHO: World Health Organization; WHO-CIDI: WHO-composite international diagnostic interview; WHO-GPAQ: WHO global physical activity questionnaire.

#### Supplementary Information

The online version contains supplementary material available at https://doi.org/10.1186/s40337-022-00644-4.

Additional file 1: Table S1. M-Box test of equality of covariance matrices for the overall difference in EAT-26 and BESSA scores of the individual and team athlete groups. **Table S2.** Levene's test of homogeneity of variance for the EAT-26 and BESSA subscale scores of the individual and team athlete groups. **Table S3.** MANOVA for the overall difference in EAT-26 and BESSA scores of the individual and team athlete groups

#### Acknowledgements

The authors wish to thank Alireza Homayouni, Ph.D., for serving as the statistical expert for this research, Heather Hower, M.S.W., for preparing the manuscript, the study participants for their contributions to this research, and the associated study research team.

#### Author contributions

MHF contributed to the conceptualization, data curation, investigation, methodology, project administration, resources, supervision, validation, visualization, and writing (reviewing and editing) of the study. SS contributed to the data curation, investigation, project administration, resources, supervision, writing (original draft, reviewing, and editing) of the study. AH contributed to the data curation, formal analysis, investigation, project administration, resources, supervision, validation, and writing (reviewing and editing) of the study. HH contributed to the project administration, resources, and writing (reviewing and editing) of the study. All authors read and approved the final manuscript.

#### Funding

This research did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors.

#### Availability of data and materials

The datasets generated and analyzed during the current study are not publicly available, as individual privacy could be compromised, but are available from the corresponding author on reasonable request.

#### Declarations

#### Ethics approval and consent to participate

The governing Medical Ethics and History of Medicine Research Center for each Iranian university study site reviewed and approved the study before enrollment. Participants were recruited via a study research coordinator who had contacted the respective officials of the 6 sports groups. The study procedures were explained, and informed consent was obtained from all participants and their parents prior to study initiation.

#### **Consent for publication**

Not applicable.

#### Competing interests

The authors declare that they have no competing interests.

#### Author details

<sup>1</sup>Department of Physical Education, Farhangian University, Tarbiat Moalem Street, Tehran, Iran. <sup>2</sup>Department of Sport Science, Faculty of Humanities, Tarbiat Modares University, Jalal Al Ahmad Street, Tehran, Iran. <sup>3</sup>Department of Psychology, Bandargaz Branch, Islamic Azad University, Bandar-e-Gaz Street, Bandargaz, Iran. <sup>4</sup>Department of Psychiatry, Eating Disorders Center for Treatment and Research, University of California at San Diego School of Medicine, 4510 Executive Drive, San Diego, CA 92121, USA. <sup>5</sup>Department of Health Services, Policy, and Practice, Hassenfeld Child Innovation Institute, Brown University School of Public Health, 121 South Main Street, Providence, RI 02903, USA.

#### Received: 4 April 2022 Accepted: 31 July 2022 Published online: 12 August 2022

#### References

- Organization WH. Infection prevention and control during health care when novel coronavirus (nCoV) infection is suspected: interim guidance, 25 January 2020. Infection prevention and control during health care when novel coronavirus (nCoV) infection is suspected: interim guidance, 25 January 2020.
- Organization WH. Laboratory testing of 2019 novel coronavirus (2019nCoV) in suspected human cases: interim guidance, 17 January 2020. Laboratory testing of 2019 novel coronavirus (2019-nCoV) in suspected human cases: interim guidance, 17 January 20202.
- Holmes EA, O'Connor RC, Perry VH, Tracey I, Wessely S, Arseneault L, et al. Multidisciplinary research priorities for the COVID-19 pandemic: a call for action for mental health science. Lancet Psychiatry. 2020;7(6):547–60.
- McGuine TA, Biese KM, Petrovska L, Hetzel SJ, Reardon CL, Kliethermes S, et al. Changes in the health of adolescent athletes: a comparison of health measures collected before and during the CoVID-19 pandemic. J Athl Train. 2021;56(8):836–44.
- Denerel N, Şenışık S, Köyağasıoğlu O, Çiğdem S, Tunç S. Effects of longduration home isolation linked to the COVID-19 pandemic on mental health of adolescent athletes. Pediatr Exerc Sci. 2021;33(4):170–6.
- 6. Rajapakse N, Dixit D. Human and novel coronavirus infections in children: a review. Pediatr Int Child Health. 2021;41(1):36–55.
- Fitzgerald HT, Rubin ST, Fitzgerald DA, Rubin BK. Covid-19 and the impact on young athletes. Paediatr Respir Rev. 2021;39:9–15.
- Buckley GL, Hall LE, Lassemillante A-CM, Belski R. Disordered eating & body image of current and former athletes in a pandemic; a convergent mixed methods study-What can we learn from COVID-19 to support athletes through transitions? J Eat Disord. 2021;9(1):1–16.
- Malek Rivan NF, Yahya HM, Shahar S, Ajit Singh DK, Ibrahim N, Mat Ludin AF, et al. The impact of poor nutrient intakes and food insecurity on the psychological distress among community-dwelling middle-aged and older adults during the COVID-19 pandemic. Nutrients. 2021;13(2):353.
- Graupensperger S, Benson AJ, Kilmer JR, Evans MB. Social (un) distancing: teammate interactions, athletic identity, and mental health of student-athletes during the COVID-19 pandemic. J Adolesc Health. 2020;67(5):662–70.
- 11. Pété E, Leprince C, Lienhart N, Doron J. Dealing with the impact of the COVID-19 outbreak: Are some athletes' coping profiles more adaptive than others? Eur J Sport Sci. 2022;22(2):237–47.
- McGuine TA, Biese KM, Petrovska L, Hetzel SJ, Reardon C, Kliethermes S, et al. Mental health, physical activity, and quality of life of US adolescent athletes during COVID-19–related school closures and sport cancellations: a study of 13 000 athletes. J Athl Train. 2021;56(1):11–9.
- Weinbach N, Lock J, Bohon C. Superior response inhibition to highcalorie foods in adolescents with anorexia nervosa. Behav Res Ther. 2020;124:103441.
- Andreato LV, Coimbra DR, Andrade A. Challenges to athletes during the home confinement caused by the COVID-19 pandemic. Strength Cond J. 2020;42:1–5.
- Gabriel BM, Zierath JR. The limits of exercise physiology: from performance to health. Cell Metab. 2017;25(5):1000–11.
- Quinn AM, Fallon BJ. Predictors of recovery time. J Sport Rehabil. 2000;9(1):62–76.
- Joy E, Kussman A, Nattiv A. 2016 update on eating disorders in athletes: a comprehensive narrative review with a focus on clinical assessment and management. Br J Sports Med. 2016;50(3):154–62.
- APA. Diagnostic and statistical manual of mental disorders: Dsm-5. Washington: American Psychiatric Publishing Incorporated; 2013.
- Maister L, De Beukelaer S, Longo M, Tsakiris M. The self in the mind's eye: reverse-correlating one's self reveals how psychological beliefs and attitudes shape our body-image. PsyArXiv Prepr; 2020.

- 21. Parks PSM, Read MH. Adolescent male athletes: body image, diet, and exercise. Adolescence. 1997;32(127):593–603.
- Gordon PA, Overbey JA. Unconscious Bias in the COVID-19 Era. In: Multidisciplinary approach to diversity and inclusion in the COVID-19-Era workplace. IGI Global; 2022. p. 158–75.
- 23. Pikoos TD, Buzwell S, Sharp G, Rossell SL. The Zoom effect: exploring the impact of video calling on appearance dissatisfaction and interest in aesthetic treatment during the COVID-19 pandemic. Aesthetic Surgery Journal. 2021;41(12):NP2066–75.
- Schneider J, Pegram G, Gibson B, Talamonti D, Tinoco A, Craddock N, et al. A mixed-studies systematic review of the experiences of body image, disordered eating, and eating disorders during the COVID-19 pandemic. Int J Eat Disord; 2022.
- Schlegl S, Maier J, Meule A, Voderholzer U. Eating disorders in times of the COVID-19 pandemic—Results from an online survey of patients with anorexia nervosa. Int J Eat Disord. 2020;53(11):1791–800.
- 26. Swami V, Horne G, Furnham A. COVID-19-related stress and anxiety are associated with negative body image in adults from the United Kingdom. Personality Individ Differ. 2021;170:110426.
- White III J. Examining physical activity behaviors and psychometric beliefs of college students before and after the COVID-19 shutdown. 2021.
- Vall-Roqué H, Andrés A, Saldaña C. The impact of COVID-19 lockdown on social network sites use, body image disturbances and self-esteem among adolescent and young women. Prog Neuropsychopharmacol Biol Psychiatry. 2021;110:110293.
- Arafa A, Shehata A, Youssef M, Senosy S. Violence against healthcare workers during the COVID-19 pandemic: a cross-sectional study from Egypt. Arch Environ Occup Health. 2021;77:621–7.
- Brewer BW, Van Raalte JL, Linder DE. Athletic identity: Hercules' muscles or Achilles heel? Int J Sport Psychol. 1993.
- Sanders G, Stevinson C. Associations between retirement reasons, chronic pain, athletic identity, and depressive symptoms among former professional footballers. Eur J Sport Sci. 2017;17(10):1311–8.
- 32. Maister L, De Beukelaer S, Longo M, Tsakiris M. The Self in the Mind's Eye: Reverse-correlating one's self reveals how psychological beliefs and attitudes shape our body-image. 2020.
- Touyz S, Lacey H, Hay P. Eating disorders in the time of COVID-19. J Eat Disord. 2020;8:19.
- Linardon J, Messer M, Rodgers RF, Fuller-Tyszkiewicz M. A systematic scoping review of research on COVID-19 impacts on eating disorders: a critical appraisal of the evidence and recommendations for the field. Int J Eat Disord. 2022;55(1):3–38.
- Graell M, Morón-Nozaleda MG, Camarneiro R, Villaseñor Á, Yáñez S, Muñoz R, et al. Children and adolescents with eating disorders during COVID-19 confinement: difficulties and future challenges. Eur Eat Disord Rev. 2020;28(6):864–70.
- Spettigue W, Obeid N, Erbach M, Feder S, Finner N, Harrison ME, et al. The impact of COVID-19 on adolescents with eating disorders: a cohort study. J Eat Disord. 2021;9(1):1–8.
- Haghshomar M, Shobeiri P, Brand S, Rossell SL, Akhavan Malayeri A, Rezaei N. Changes of symptoms of eating disorders (ED) and their related psychological health issues during the COVID-19 pandemic: a systematic review and meta-analysis. J Eat Disord. 2022;10(1):1–18.
- Touyz S, Lacey H, Hay P. Eating disorders in the time of COVID-19. Berlin: Springer; 2020. p. 1–3.
- 39. Petrie TA, Greenleaf CA. Eating disorders in sport: from theory to research to intervention. 2007.
- Dockendorff SA, Petrie TA, Greenleaf CA, Martin S. Intuitive Eating Scale: An examination among early adolescents. J Couns Psychol. 2012;59(4):604.
- Martinsen M, Bratland-Sanda S, Eriksson AK, Sundgot-Borgen J. Dieting to win or to be thin? A study of dieting and disordered eating among adolescent elite athletes and non-athlete controls. Br J Sports Med. 2010;44(1):70–6.
- Kong P, Harris LM. The sporting body: body image and eating disorder symptomatology among female athletes from leanness focused and nonleanness focused sports. J Psychol. 2015;149(2):141–60.

- Mountjoy M, Sundgot-Borgen J, Burke L, Carter S, Constantini N, Lebrun C, et al. The IOC consensus statement: beyond the female athlete triad—relative energy deficiency in sport (RED-S). Br J Sports Med. 2014;48(7):491–7.
- Mancine RP, Gusfa DW, Moshrefi A, Kennedy SF. Prevalence of disordered eating in athletes categorized by emphasis on leanness and activity type–a systematic review. J Eat Disord. 2020;8(1):1–9.
- 45. Andersen MB. Sport psychology in practice: Human kinetics; 2005.
- 46. McFee G. Officiating in aesthetic sports. J Philos Sport. 2013;40(1):1-17.
- Krentz E, Warschburger P. A longitudinal investigation of sports-related risk factors for disordered eating in aesthetic sports. Scand J Med Sci Sports. 2013;23(3):303–10.
- Krentz EM, Warschburger P. Sports-related correlates of disordered eating in aesthetic sports. Psychol Sport Exerc. 2011;12(4):375–82.
- Giel KE, Hermann-Werner A, Mayer J, Diehl K, Schneider S, Thiel A, et al. Eating disorder pathology in elite adolescent athletes. Int J Eat Disord. 2016;49(6):553–62.
- Chatterton JM, Petrie TA. Prevalence of disordered eating and pathogenic weight control behaviors among male collegiate athletes. Eat Disord. 2013;21(4):328–41.
- Weber SE, Harris MM, Wright HH, Manore MM. Assessment of disordered eating and orthorexia nervosa in endurance athletes following gluten and wheat-free diets (GWD). Med Sci Sports Exerc. 2017;49(55):712.
- 52. Ogden J, Roy-Stanley C. How do children make food choices? Using a think-aloud method to explore the role of internal and external factors on eating behaviour. Appetite. 2020;147:104551.
- Landkammer F, Winter K, Thiel A, Sassenberg K. Team sports off the field: Competing excludes cooperating for individual but not for team athletes. Front Psychol. 2019;10:2470.
- Stice E, Marti CN, Rohde P. Prevalence, incidence, impairment, and course of the proposed DSM-5 eating disorder diagnoses in an 8-year prospective community study of young women. J Abnorm Psychol. 2013;122(2):445.
- Allen KL, Byrne SM, Oddy WH, Crosby RD. DSM–IV–TR and DSM-5 eating disorders in adolescents: prevalence, stability, and psychosocial correlates in a population-based sample of male and female adolescents. J Abnorm Psychol. 2013;122(3):720.
- Smink FR, van Hoeken D, Oldehinkel AJ, Hoek HW. Prevalence and severity of DSM-5 eating disorders in a community cohort of adolescents. Int J Eat Disord. 2014;47(6):610–9.
- Weltzin TE, Weisensel N, Franczyk D, Burnett K, Klitz C, Bean P. Eating disorders in men: Update. J Men's Health Gend. 2005;2(2):186–93.
- Dearden A, Mulgrew KE. Service provision for men with eating issues in Australia: an analysis of organisations', practitioners', and men's experiences. Aust Soc Work. 2013;66(4):590–606.
- Muise AM, Stein DG, Arbess G. Eating disorders in adolescent boys: a review of the adolescent and young adult literature. J Adolesc Health. 2003;33(6):427–35.
- Stoyel H, Slee A, Meyer C, Serpell L. Systematic review of risk factors for eating psychopathology in athletes: a critique of an etiological model. Eur Eat Disord Rev. 2020;28(1):3–25.
- Milligan B, Pritchard M. The relationship between gender, type of sport, body dissatisfaction, self esteem and disordered eating behaviors in division I athletes. Athletic Insight. 2006;8(1):32–46.
- 62. Limbers CA, Cohen LA, Gray BA. Eating disorders in adolescent and young adult males: prevalence, diagnosis, and treatment strategies. Adolesc Health Med Ther. 2018;9:111.
- Rice FP. Human development: A life-span approach. Hoboken: Prentice Hall; 1998.
- Organization WH. Global physical activity questionnaire (GPAQ) analysis guide. Geneva: World Health Organization; 2012. p. 1–22.
- Kessler RC, Üstün TB. The world mental health (WMH) survey initiative version of the world health organization (WHO) composite international diagnostic interview (CIDI). Int J Methods Psychiatr Res. 2004;13(2):93–121.
- Chuan CL, Penyelidikan J. Sample size estimation using Krejcie and Morgan and Cohen statistical power analysis: a comparison. J Penyelidikan IPBL. 2006;7(1):78–86.
- 67. Bull FC, Maslin TS, Armstrong T. Global physical activity questionnaire (GPAQ): nine country reliability and validity study. J Phys Act Health. 2009;6(6):790–804.

- Craig CL, Marshall AL, Sjöström M, Bauman AE, Booth ML, Ainsworth BE, et al. International physical activity questionnaire: 12-country reliability and validity. Med Sci Sports Exerc. 2003;35(8):1381–95.
- Esteghamati A, Khalilzadeh O, Rashidi A, Kamgar M, Meysamie A, Abbasi M. Physical activity in Iran: results of the third national surveillance of risk factors of non-communicable diseases (SuRFNCD-2007). J Phys Act Health. 2011;8(1):27–35.
- Lawshe CH. A quantitative approach to content validity. Pers Psychol. 1975;28(4):563–75.
- Dadfar M, Momeni Safarabad N, Asgharnejad Farid AA, Nemati Shirzy M, Ghazie Pour Abarghouie F. Reliability, validity, and factorial structure of the World Health Organization-5 Well-Being Index (WHO-5) in Iranian psychiatric outpatients. Trends Psychiatry Psychother. 2018;40:79–84.
- Spitzer RL, Kroenke K, Williams JB, Group PHQPCS, Group PHQPCS. Validation and utility of a self-report version of PRIME-MD: the PHQ primary care study. Jama. 1999;282(18):1737–44.
- Kroenke K, Spitzer RL, Williams JB. The PHQ-15: validity of a new measure for evaluating the severity of somatic symptoms. Psychosom Med. 2002;64(2):258–66.
- 74. Beck AT, Beck RW. Screening depressed patients in family practice: a rapid technic. Postgrad Med. 1972;52(6):81–5.
- Davari Ashtiani R, Alaghbandrad J, Sharifi V, Amini H, Kaviani H, Shaabani A, et al. Reliability of the persian translation of the composite international diagnostic interview (Cidi) for diagnosing schizophrenia and bipolar disorder. Adv Cognit Sci. 2004;6(1):1–9.
- Garner D, Olmsted M, Bohr Y, Garfinkel P. The eating attitudes test: psychometric features. Psychol Med. 1982;12:871–8.
- 77. George D, Mallery P. edition 4. SPSS for Windows step by step: A simple guide and reference. 11.0 update. Boston. Allyn & Bacon; 2003.
- Gleaves DH, Pearson CA, Ambwani S, Morey LC. Measuring eating disorder attitudes and behaviors: a reliability generalization study. J Eat Disord. 2014;2(1):1–12.
- First MB, Spitzer RL, Gibbon M, Williams JB. Structured clinical interview for Axis I DSM-IV disorders. New York: Biometrics Research; 1994.
- Mintz LB, O'Halloran MS. The eating attitudes test: validation with DSM-IV eating disorder criteria. J Pers Assess. 2000;74(3):489–503.
- Garner DM, Olmsted MP, Bohr Y, Garfinkel PE. The eating attitudes test: psychometric features and clinical correlates. Psychol Med. 1982;12(4):871–8.
- Lee S, Kwok K, Liau C, Leung T. Screening Chinese patients with eating disorders using the eating attitudes test in Hong Kong. Int J Eat Disord. 2002;32(1):91–7.
- Sahlan RN, Blomquist KK, Bodell LP. Psychometric properties of the Farsi version of the eating pathology symptoms inventory (F-EPSI) among Iranian University men and women. J Eat Disord. 2022;10(1):1–9.
- Mendelson BK, Mendelson MJ, White DR. Body-esteem scale for adolescents and adults. J Pers Assess. 2001;76(1):90–106.
- Pazhoohi F, Hosseinchari M, Doyle J. Iranian men's waist-to-hip ratios, shoulder-to-hip ratios, body esteem and self-efficacy. J Evol Psychol. 2012;10(2):61–7.
- Released IC. IBM SPSS Statistics for Windows, Version 24.0. Armonk, NY: IBM Corp. 2016.
- Steinskog DJ, Tjøstheim DB, Kvamstø NG. A cautionary note on the use of the Kolmogorov-Smirnov test for normality. Mon Weather Rev. 2007;135(3):1151–7.
- French A, Macedo M, Poulsen J, Waterson T, Yu A. Multivariate analysis of variance (MANOVA). 2008.
- Friendly M, Sigal M. Visualizing tests for equality of covariance matrices. Am Stat. 2020;74(2):144–55.
- Nordstokke DW, Zumbo BD. A new nonparametric Levene test for equal variances. Psicológica. 2010;31(2):401–30.
- 91. Prevention CfDCCa. BMI Percentile Calculator for Child and Teen. 2021. Available from: https://www.google.com/url?sa=t&rct=j&q=&esrc=s& source=web&cd=&cad=rja&uact=8&ved=2ahUKEwi3sev83sX2AhW Ahv0HHSbNCO8QFnoECAYQAQ&url=https%3A%2F%2Fwvw.cdc.gov% 2Fhealthyweight%2Fbmi%2Fcalculator.html&usg=AOvVaw0mQCuKklh 9J7rOkQBdYs7m.
- 92. Pamuk Ö, Makaracı Y, Özer Ö, Soslu R. Examination of eating attitudes of Universities' Sports Department Students In Terms of Individual and Team Sports. Int J Appl Exerc Physiol. 2020;9(12):43–9.

- Jm MN, Jp NR, Jiménez R. Physical exercise and food habits: a study of adolescents in Cadiz. Rev Esp Salud Publica. 1999;73(1):81–7.
- 94. Heradstveit O, Hysing M, Nilsen SA, Bøe T. Symptoms of disordered eating and participation in individual-and team sports: a population-based study of adolescents. Eat Behav. 2020;39:101434.
- 95. Frieiro P, González-Rodríguez R, Domínguez-Alonso J. Self-esteem and socialisation in social networks as determinants in adolescents' eating disorders. Health & Social Care in the Community. 2022.
- Kullmann S, Giel KE, Hu X, Bischoff SC, Teufel M, Thiel A, et al. Impaired inhibitory control in anorexia nervosa elicited by physical activity stimuli. Soc Cognit Affect Neurosci. 2014;9(7):917–23.

## **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

#### Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

#### At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

